Author’s response to reviews

Title: Effects of adding psychosocial stimulation for children of lactating mothers using an unconditional cash transfer platform on neurocognitive behavior of children in rural Bangladesh: protocol for a cluster randomized controlled trial

Authors:
Sheikh Hossain (sheikh.jamal@icddrb.org)
Bharati Roy (bharati.roy@icddrb.org)
Nur-E- Salveen (n.salveen@icddrb.org)
Mohammad Hasan (imrul@icddrb.org)
S.M. Mulk Tipu (tipu.mulk@icddrb.org)
Shamima Shiraji (shamima.shiraji@icddrb.org)
Fahmida Tofail (ftofail@icddrb.org)
Jena Hamadani (jena@icddrb.org)

Version: 1 Date: 24 Oct 2018

Author’s response to reviews:

Editor's comments and our responses:

1. Editor comments: Ethical and Funding Approval Documentation

Before we can proceed with your submission, can you please forward copies of all ethical approval and funding approval for our records. These documents should be sent as email attachments to the following email address, BMCSeriesEditorial@biomedcentral.com. Please DO NOT upload these documents as additional files in the submission system. If your documents are not in English, please could you provide translated versions of the relevant parts. These should be endorsed and signed by a contactable person at the institution. Please also include the original documents. Can you please confirm in your cover letter that you have forwarded the requested documentation to BMCSeriesEditorial@biomedcentral.com.

Answer: We have already sent it
2. Funding

A study is considered to be externally funded if the authors have been awarded a grant for the study by a major funding body (e.g. governmental funding/award from a charitable foundation). If a study has not received external funding, then the study protocol will be sent for peer-review with a member of our Editorial Board. If a study has received funding/assistance from a commercial organization, this should be clearly stated in the 'competing interests' section of your manuscript, and the study protocol will be sent for peer-review by a member of our Editorial Board. Can you please confirm whether your study protocol has undergone peer-review by the funding body?

Answer: The donor of this funding is Grand Challenges Canada and the donor took funding decision after peer review.

3. The order, information, and section titles of your manuscript are currently incorrect. Please read Submission Guidelines carefully to make sure the formatting, section order, and relevant information are all correct. https://bmcpsychology.biomedcentral.com/submission-guidelines/preparing-your-manuscript/study-protocol.

Answer: We have corrected the file as per your guide.

We are waiting for your quick decision as there are a number of implications for rapid decision.

Thank you

Sheikh Jamal Hossain

Thank you